Author's response to reviews

Title: Blood cultures for the diagnosis of multidrug-resistant and extensively drug-resistant tuberculosis among HIV-infected patients from rural South Africa: a cross-sectional study

Authors:

Scott K Heysell (scott.heysell@gmail.com)
Tania A Thomas (tat3x@virginia.edu)
Neel R Gandhi (neelrgandhi@gmail.com)
Francois J Eksteen (fjeksteen@gmail.com)
Anthony P Moll (4tonymoll@gmail.com)
Yacoob Coovadia (coovadia@ukzn.ac.za)
Lynette Roux (LynetteRou@ialch.co.za)
Palav Babaria (p.babaria@gmail.com)
Umesh Laloo (lalloo@ukzn.ac.za)
Gerald Friedland (gerald.friedland@yale.edu)
Sarita Shah (saritashah2@gmail.com)

Version: 3 Date: 2 December 2010

Author's response to reviews:

December 01, 2010

Dear Editors,

We thank you and the reviewers for your acceptance of the manuscript # 1207148427406711, “Blood cultures for the diagnosis of multidrug-resistant and extensively drug-resistant tuberculosis among HIV-infected patients from rural South Africa: a cross-sectional study.” We have provided the requested formatting changes.

Specifically, we have formatted the author list in style and order. All track changes have been removed. Reference #5 is now provided in full. The vertical lines of the tables and the merged cells of the table were removed.

Please let us know if we can make any further adjustments.

Sincerely,
Scott K. Heysell MD, MPH
University of Virginia